Author's response to reviews

Title: Development of a prototype clinical decision support tool for osteoporosis disease management: A qualitative study of focus groups

Authors:

- Monika Kastner (monika.kastner@utoronto.ca)
- Jamy Li (jamy.li@utoronto.ca)
- Danielle Lottridge (danielle.lottridge@utoronto.ca)
- Christine Marquez (cmarquez@utoronto.ca)
- David Newton (david.newton@utoronto.ca)
- Sharon E Straus (sharon.straus@utoronto.ca)

Version: 3 Date: 19 July 2010

Author's response to reviews: see over
Response to Reviewers

Development of a Prototype Clinical Decision Support Tool for Osteoporosis Disease Management: A Qualitative Study of Focus Groups

Dear Editor:

Thank you very much for the opportunity to publish our work in your journal. We made the following edits and corrections to the manuscript:

- Author list corrected
- Table shading removed
- Table labeling and listing corrected
- Referencing changed to square brackets
- Renamed Appendix to “Additional files”
- Changed sentence case in the title and submission system
- Included country in the Affiliations
- Removed tracking from the manuscript
- Removed underlining from the manuscript
- Moved statement of funding to Acknowledgements
- Removed merged cells from table
- Removed ranges form listing figures
- References contain all authors names